Author’s response to reviews

Title: Fascia iliaca compartment block as a preoperative analgesic in elderly patients with hip fractures – effects on cognition. A prospective randomised trial.

Authors:

Pär Wennberg (par_wennberg@hotmail.com)
Margareta Möller (margareta.moller@regionorebrolan.se)
Johan Herlitz (johan.herlitz@hb.se)
Elisabeth Kenne Sarenmalm (elisabeth.kenne.sarenmalm@vgregion.se)

Version: 1 Date: 17 Feb 2019

Author’s response to reviews:

Before it can be sent out for review, please carry out the corrections, below:

-Provide the header "Declarations"
Action taken: Header "Declarations" inserted

- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Action taken: A text is added regarding funding."This study received support from Örebro University Hospital and Skaraborg Hospital as PW received salary during design of the study and collection, analysis, and interpretation of data and in writing the manuscript."

- In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
Action taken: The following text is added. "Written consent was obtained from participants. Patients who were assessed as not having the capacity for consent at the time of inclusion, were included following presumed consent; this was supported by decision from the Regional Ethics Board in Uppsala, as supported by Swedish law. This assessment was made by the including physician, together with the nurse responsible for the patient. The Short Portable Mental Status Questionnaire was used to support the decision of inclusion on presumed consent."

- Your availability of data and materials section is insufficient. Please de-identify the data and
allow your raw data to be shared, replacing your phrase with: The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.

Action taken: The following text is added. "The datasets used and/or analysed during the current study available from the corresponding author on reasonable request."

- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Action taken: A clean revised manuscript is uploaded.